Author's response to reviews

Title: The Translation Research in a Dental Setting (TRiaDS) Programme Protocol.

Authors:

Jan E Clarkson (j.e.clarkson@cpse.dundee.ac.uk)
Craig R Ramsay (c.r.ramsay@abdn.ac.uk)
Martin P Eccles (martin.eccles@ncl.ac.uk)
Sandra Eldridge (s.eldridge@qmul.ac.uk)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Marie Johnston (m.johnston@abdn.ac.uk)
Susan Michie (s.michie@ucl.ac.uk)
Shaun Treweek (s.treweek@cpse.dundee.ac.uk)
Alan Walker (Alan.Walker@nes.scot.nhs.uk)
Linda Young (Linda.Young@nes.scot.nhs.uk)
Irene Black (Irene.Black@nes.scot.nhs.uk)
Debbie Bonetti (d.bonetti@cpse.dundee.ac.uk)
Heather Cassie (h.cassie@cpse.dundee.ac.uk)
Jill Francis (j.francis@abdn.ac.uk)
Gillian MacKenzie (gillian.mackenzie@nes.scot.nhs.uk)
Lorna MacPherson (l.macpherson@dental.gla.ac.uk)
Lorna McKee (l.mckee@abdn.ac.uk)
Nigel Pitts (n.b.pitts@cpse.dundee.ac.uk)
Jim Rennie (Jim.Rennie@nes.scot.nhs.uk)
Doug Stirling (douglas.stirling@nes.scot.nhs.uk)
Colin Tilley (c.tilley@nhs.net)
Carole Torgerson (cjt3@york.ac.uk)
Luke Vale (l.vale@abdn.ac.uk)

Version: 4 Date: 22 June 2010

Author's response to reviews:

22 June 2010

Dear Editors

Re manuscript The Translation Research in a Dental Setting (TRiaDS) Programme Protocol (Jan E Clarkson et al MSID 6057842783454396)

We wish to submit the above manuscript for formal acceptance and publication in Implementation Science.

As per your instructions, the manuscript has been checked and the changes have been accepted.
The manuscript contains the abstract, table 1, 2, 3 and 4 and Figure 1.

We have made one amendment to the main document on page 14 – the last word at the end of paragraph one has been changed from analysed to considered as we feel that this word choice is more appropriate.

Carole Torgerson, an author on this manuscript, no longer works for the University of York and has joined the University of Birmingham. This institution change has been noted at the end of the author page, as per your email instructions.

Please contact me if you would like any further information. We look forward to your response to this submission.

Yours sincerely

Jan Clarkson (on behalf of all authors)